Author's response to reviews

Title: DMBT1 expression is down-regulated in breast cancer

Authors:

Paola P Braidotti (paola.braidotti@unimi.it)
Paolo P Nuciforo (nuciforo@ifom-firc.it)
Jan J Mollenhauer (j.mollenhauer@dkfz-heidelberg.de)
Annemarie A Poustka (a.poustka@dkfz-heidelberg.de)
Caterina C Pellegrini (catepeita@yahoo.com)
Alessia A Moro (alessiamoro@tiscali.it)
Gaetano G Bulfamante (gaetano.bulfamante@unimi.it)
Guido G Coggi (guido.coggi@unimi.it)
Silvano S Bosari (silvano.bosari@unimi.it)
Giuseppe G Pietra (gkpietra@freesurf.ch)

Version: 4 Date: 9 August 2004

Author's response to reviews:

Reviewer 2 suggested that figure 3 should be present in higher magnification. This could be obtained digitally by the readers of the Journals. Anyway we inserted an inset with a higher magnification in the left upper corners of figure 3 a, b, and c, preserving the original magnification in order to allow a correct identification of the same section area.